

GOPEN ACCESS

Citation: Hoskens H, Liu D, Naqvi S, Lee MK, Eller RJ, Indencleef K, et al. (2021) 3D facial phenotyping by biometric sibling matching used in contemporary genomic methodologies. PLoS Genet 17(5): e1009528. https://doi.org/10.1371/ journal.pgen.1009528

Editor: Sijia Wang, Shanghai Institutes for Biological Sciences, CHINA

Received: July 28, 2020

Accepted: April 1, 2021

Published: May 13, 2021

Copyright: © 2021 Hoskens et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All of the genotypic markers for the PITT dataset are available to the research community through the dbGaP controlled access repository (http://www.ncbi.nlm.nih.gov/ gap) at accession number phs000949.v1.p1. The raw source data for the phenotypes -the 3D facial surface models in.obj format- are available through the FaceBase Consortium (www.facebase.org). Access to these 3D facial surface models requires proper institutional ethics approval and approval from the FaceBase data access committee. The participants making up the family-based (SIB), RESEARCH ARTICLE

3D facial phenotyping by biometric sibling matching used in contemporary genomic methodologies

Hanne Hoskens^{1,2*}, Dongjing Liu³, Sahin Naqvi^{4,5}, Myoung Keun Lee⁶, Ryan J. Eller⁷, Karlijne Indencleef^{2,8,9}, Julie D. White¹⁰, Jiarui Li^{2,8}, Maarten H. D. Larmuseau^{1,11,12}, Greet Hens⁹, Joanna Wysocka^{4,13,14}, Susan Walsh⁷, Stephen Richmond¹⁵, Mark D. Shriver¹⁰, John R. Shaffer^{3,6}, Hilde Peeters¹, Seth M. Weinberg^{3,6,16}, Peter Claes^{1,2,8,17}*

1 Department of Human Genetics, KU Leuven, Leuven, Belgium, 2 Medical Imaging Research Center, UZ Leuven, Leuven, Belgium, 3 Department of Human Genetics, University of Pittsburgh, Pittsburgh, Pennsylvania, United States of America, 4 Department of Chemical and Systems Biology, Stanford University School of Medicine, Stanford, California, United States of America, 5 Department of Genetics, Stanford University School of Medicine, Stanford, California, United States of America, 6 Department of Oral Biology, Center for Craniofacial and Dental Genetics, University of Pittsburgh, Pittsburgh, Pennsylvania, United States of America, 7 Department of Biology, Indiana University Purdue University Indianapolis, Indianapolis, Indiana, United States of America, 8 Department of Electrical Engineering, ESAT/PSI, KU Leuven, Leuven, Belaium, 9 Department of Otorhinolaryngology, KU Leuven, Leuven, Belaium, 10 Department of Anthropology, The Pennsylvania State University, State College, Pennsylvania, United States of America, 11 Department of Biology, Laboratory of Socioecology and Social Evolution, KU Leuven, Leuven, Belgium, 12 Histories vzw, Mechelen, Belgium, 13 Department of Developmental Biology, Stanford University School of Medicine, Stanford, California, United States of America, 14 Howard Hughes Medical Institute, Stanford University School of Medicine, Stanford, California, United States of America, 15 Applied Clinical Research and Public Health, School of Dentistry, Cardiff University, Cardiff, United Kingdom, 16 Department of Anthropology, University of Pittsburgh, Pittsburgh, Pennsylvania, United States of America, 17 Murdoch Children's Research Institute, Melbourne, Victoria, Australia

* hanne.hoskens@kuleuven.be (HH); peter.claes@kuleuven.be (PC)

Abstract

The analysis of contemporary genomic data typically operates on one-dimensional phenotypic measurements (e.g. standing height). Here we report on a data-driven, family-informed strategy to facial phenotyping that searches for biologically relevant traits and reduces multivariate 3D facial shape variability into amendable univariate measurements, while preserving its structurally complex nature. We performed a biometric identification of siblings in a sample of 424 children, defining 1,048 sib-shared facial traits. Subsequent quantification and analyses in an independent European cohort (n = 8,246) demonstrated significant heritability for a subset of traits (0.17–0.53) and highlighted 218 genome-wide significant loci (38 also study-wide) associated with facial variation shared by siblings. These loci showed preferential enrichment for active chromatin marks in cranial neural crest cells and embryonic craniofacial tissues and several regions harbor putative craniofacial genes, thereby enhancing our knowledge on the genetic architecture of normal-range facial variation.

Penn State University (PSU) and Indiana University-Purdue University Indianapolis (IUPUI) datasets were not collected with broad data sharing consent. Given the highly identifiable nature of both facial and genomic information and unresolved issues regarding risks to participants of inherent reidentification, participants were not consented for inclusion in public repositories or the posting of individual data. This restriction is not because of any personal or commercial interests. Further information about access to the raw 3D facial images and/or genomic data can be obtained from the respective ethics committees; the Ethics Committee Research UZ/KU Leuven (ec@uzleuven. be), the PSU IRB (IRB-ORP@psu.edu), and the IUPUI IRB (irb@iu.edu) for the SIB, PSU and IUPUI datasets, respectively. The ALSPAC (UK) data will be made available to bona fide researchers on application to the ALSPAC Executive Committee (http://www.bris.ac.uk/alspac/researchers/dataaccess). Ethical approval for the study was obtained from the ALSPAC Ethics and Law Committee and the Local Research Ethics Committees. KU Leuven provides the MeshMonk spatially dense facial mapping software, free to use for academic purposes (https://github.com/ TheWebMonks/meshmonk). Matlab implementations of the hierarchical spectral clustering to obtain facial segmentations are available from a previous publication [29] (10. 6084/m9.figshare.7649024). Meta-analysis GWAS statistics are available on GWAS Catalog (study accession GCST9001514-GCST90016561). All

relevant data to run future replications and metaanalysis efforts are provided in Matlab format online (https://figshare.com/s/ 53f56ed70c76cda807a8) [30]. This includes the anthropometric mask used, facial segmentation cluster labels, PCA shape spaces for all 63 facial segments, sibling-derived trait vectors and

association statistics for the lead SNPs.

Funding: The KU Leuven research team (P.C., H.P., G.H.) and analyses were supported by the National Institutes of Health (1-R01-DE027023), The Research Fund KU Leuven (BOF-C1, C14/15/081) and The Research Program of the Research Foundation - Flanders (Belgium) (FWO, G078518N). The collaborators at the University of Pittsburgh (S.M.W., J.R.S.) were supported by the National Institute of Dental and Craniofacial Research (U01-DE020078, R01-DE016148, and R01-DE027023). Funding for genotyping by the National Human Genome Research Institute (X01-HG007821 and X01-HG007485) and funding for initial genomic data cleaning by the University of Washington provided by contract HHSN268201200008I from the National Institute

Author summary

The human face is a highly variable trait composed of distinct features, each influenced by genetic and environmental factors. The strong genetic component is primarily evidenced by the facial similarity between identical twins and the clear facial resemblances within families. Over the past decade, a powerful methodological toolbox of computational and statistical genetics has been developed to study the genetic architecture of complex traits. However, these methods typically require one-dimensional phenotypic measurements (e.g. width of the nose or spacing between the eyes) that fail to accurately describe 3D facial shape. In this paper, we learn from 3D facial data itself, a series of relevant traits that are guided by the facial similarity observed between sibling pairs. Importantly, while preserving the structural convolution of the face, these traits also fit the requirements as input to the well-established statistical tools. In doing so, we have identified many genetic loci that are associated with a wide range of facial features. Some of these regions contained genes related to embryonic facial development, and craniofacial malformations. An improved understanding of the genetic basis of facial shape can have several important applications, for example in developmental biology, medical genetics and forensic sciences.

Introduction

Systematic characterization of facial morphology is important in a variety of domains such as anthropology, medicine, and genetics [1,2]. It has the potential to provide insight into human evolutionary processes [3], to facilitate surgical planning and outcome assessment [4,5], and to guide syndrome delineation [6], among others. However, fully capturing complex multipartite traits like human 3D facial shape is not straightforward. Traditionally, this has been done using simple anthropometric measurements (e.g. linear distances, angles, and ratios) or principal components (PCs) that are derived from specific points, called landmarks, on a set of two-dimensional (2D) or three-dimensional (3D) facial images [7–16]. However, simple geometric features such as distances fail to capture the full morphological complexity of human 3D facial shape and a priori selection of traits rarely takes into account biological knowledge. As an alternative, Claes et al. [17] recently proposed an open-ended description of facial variation, thereby avoiding any preselection of individual traits. However, highly multivariate phenotypes do not lend themselves to many of the standard tools available in statistical and quantitative genetics, and the number of follow-up analyses therefore remains limited.

To address the current limitations, we propose to prioritize, in a supervised and data-driven manner, specific facial traits of interest within the multidimensional facial space, in order to reduce the complexity of 3D facial shape into genetically informed and therefore biologically relevant facial traits. In this context, families are potentially informative, as the similarities among family members clearly indicate a heritable component of facial shape [18,19]. Following Fig 1, we aimed to identify facial traits that are shared among sibling pairs in a biometric matching experiment of siblings. Subsequent scoring of a large, separate genotyped cohort for the sib-shared traits provides for the conversion of complex 3D structures into their univariate equivalents so that well-established, open-source bioinformatics tools [20–23] could be applied for further investigation. We sought to identify genetic variants contributing to variation among the sib-shared traits through genome-wide association analysis (GWAS), highlighting a combination of novel and previously identified genetic loci. We further investigated these loci in the context of early craniofacial development and morphogenesis and examined the

for Dental and Craniofacial Research awarded to the Center for Inherited Disease Research. The collaborators at the Pennsylvania State University (M.D.S.) were supported by the Center for Human Evolution and Development at Penn State, the Science Foundation of Ireland Walton Fellowship (04.W4/B643), the US National Institute of Justice (2008-DN-BX-K125 and 2018-DU-BX-0219), the US Department of Defense, and the University of Illinois Interdisciplinary Innovation Initiative Research Grant. The collaborators at the Indiana University-Purdue University Indianapolis (S.W.) were supported by the National Institute of Justice (2015-R2-CX-0023, 2014-DN-BX-K031, and 2018-DU-BX-0219). The collaborators at Stanford University (J.W.) were supported by the National Institutes of Health (1-R01-DE027023 and U01-DE024430), the Howard Hughes Medical Institute, and the March of Dimes Foundation (1-FY15-312). The UK Medical Research Council and Wellcome (Grant ref: 102215/2/13/2) and the University of Bristol provide core support for ALSPAC. The publication is the work of the authors and H.H. and P.C. will serve as guarantors for the contents of this paper. A comprehensive list of grants funding is available on the ALSPAC website http://www. bristol.ac.uk/alspac/external/documents/grantacknowledgements.pdf). ALSPAC GWAS data was generated by Sample Logistics and Genotyping Facilities at Wellcome Sanger Institute and LabCorp (Laboratory Corporation of America) using support from 23andMe. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

genetic overlap among sib-shared facial traits. Our analyses have revealed a large number of genetic variants affecting facial traits shared by siblings, illustrating their biological relevance and further enhancing our understanding of the genetic basis of human facial shape.

Results

Global-to-local facial segmentation

Facial variation was studied at multiple levels of scale by subdividing facial shape into 63 hierarchical segments. First, homologous spatially dense quasi-landmark configurations (n = 7,160) were established by the mapping of a generic template mesh onto the images [1,24]. Subsequently, facial shape was hierarchically clustered into a series of global-to-local facial segments in a data-driven manner by grouping quasi-landmarks that are strongly correlated in a large European cohort. We then applied principal component analysis (PCA) to construct a multidimensional shape space for each facial segment independently, the dimensions of which characterize facial variation [17,25]. These segments each captured different aspects of facial shape, representing the full face (segment 1), midface (segment 2) and outer face (segment 3), as well as variations in smaller regions near the philtrum (quadrant I), nose (quadrant II), lower face (quadrant III) and upper face (quadrant IV) (Fig 2A).

Biometric sibling matching

Since siblings are likely to share facial features due to close kinship, biologically relevant traits can be extracted from phenotypically similar sibling pairs. We aimed to identify these pairs and derive their overlapping facial features through a biometric identification setup, which essentially involves the comparison of facial shape between siblings in the context of multiple facial comparisons with unrelated individuals (Fig 2B). To this end, we used 3D facial images from a total of 273 sibling pairs (78 brother-brother, 79 sister-sister, 116 brother-sister) from 194 nuclear families of self-reported European ancestry (SIB cohort; S1 Table). Individual faces can be described as single points within the multidimensional space constructed per facial segment. Faces that appear to be more similar are closer together within this space, so that distance metrics can be used to measure similarity [26]. Different distance and angle-based measures were quantified, either defined in a Euclidean space (ED, EA) or Mahalanobis space (MD, MA), the latter where each dimension was weighted in terms of its variance [26,27]. In brief, the lower the distance or angle between two points, the greater the similarity between those individuals.

The ability of the different similarity measures to identify siblings was tested in a biometric identification task. In a one-to-many setup, faces were matched against a gallery of candidates for individual and combined facial segments. The rank-k% identification rate then indicates the proportion of times the true sibling was present in the top-k% matches as determined by the similarity score [28]. Therefore, matching at a low rank-k% signifies facially close-to-identical siblings. Overall matching performance of siblings was summarized via cumulative match characteristic (CMC) curves (Figs 2B and S1), where high identification rates and a steep slope of the curve at higher ranks (i.e. low rank values) indicate better performance [29]. To account for the effect of gallery size on the identification performance, results were plotted as a percentage of rank rather than absolute rank values (S2 Fig). In general, angle measurements outperformed distance measurements and performance could be further increased by adjusting for the variances of the PCs in the Mahalanobis space (S1A Fig). Hence, in this study the Mahalanobis angle (MA) was the preferred similarity measure, with the true sibling occurring within the top-1%, top-10% and top-20% candidates in 36.81%, 77.47% and 89.19% of the full-face matching experiments, respectively. While matching at the global, full-face level consistently



Fig 1. Workflow of the study.

https://doi.org/10.1371/journal.pgen.1009528.g001

performed better than individual per-segment matchers ($R1_{MA,fullface} = 36.81\%$; Fig 2B), an increase in performance was observed when all segments branching from the full face (i.e. segments 1–63) were combined ($R1_{MA,combined} = 44.32\%$; S1B Fig). The complete list of matching results for the different similarity measures and segments is provided in S2 Table.

Data-driven selection of sib-shared traits

Siblings could be matched with varying levels of accuracy depending on the similarity measure and facial features (global-to-local) being tested (Figs <u>2B</u> and <u>S1</u>). These results suggest that many siblings share one or multiple features in the face, while others look rather different. Derivation of the sib-shared traits was focused on sibling pairs that matched near perfectly (within the top-1%) in any of the segments. We defined the final trait as the average shape of a particular sibling pair within a given segment, highlighting the facial features they have in common, i.e. those that were informative for accurately matching the siblings, while masking their dissimilarities.

A total of 1,048 traits were extracted across all segments, each of them representing a particular facial feature shared by a specific pair of siblings. Visual representations of the sib-shared traits are available online [30]. The 1,048 traits comprised 322 independent traits [31] due to the hierarchical and overlapping construction of the facial segments and the presence of



Fig 2. Global-to-local matching of siblings. (A) Global-to-local segmentation of 3D facial shape obtained using hierarchical spectral clustering of the EURO cohort. Segments are colored per quadrant, represented by the roman numerals. (B) Phenotypically similar sibling pairs were identified in a biometric identification setup, which involves the comparison of facial shape between siblings and with unrelated individuals. Matching performance using different similarity measures and facial features was evaluated using cumulative match characteristic (CMC) curves. Plotted is the percentage of sibling pairs that were correctly identified (y-axis) within the top-k% matches (x-axis) using the Mahalanobis angle. Curves are colored based on the facial features that were used to match siblings. For each quadrant, the highest and lowest identification rates per rank are shown, with the area between the two shaded.

https://doi.org/10.1371/journal.pgen.1009528.g002

multiple sibling pairs per nuclear family. Facial regions that were more often shared between siblings included the orbital and nasal area as well as the mandible (Fig 3A). From the total of 273 sibling pairs, 218 pairs (60 brother-brother, 66 sister-sister, 92 brother-sister) were used at least once to define the traits, belonging to 160 out of 194 nuclear families. The remaining 55 pairs with poor matching behavior were omitted. No sex effect was observed in the matching of siblings (S3 Fig). That is, relatively equal numbers of same-sex and different-sex pairs were selected (Fisher's exact test p-value = 0.88) since sexual dimorphism was corrected for during image preprocessing. Similarly, matching of pairs was independent of the age difference of their individuals (two-sample t-test p-value = 0.87).

Reduction of 3D facial shape to univariate measurements

Sib-shared traits can be described as vectors within the multidimensional shape space, extending from the global mean shape to the average facial shape of both siblings (S4 Fig). Moving further along this direction would produce a more exaggerated version (caricature) of the sibling average, while projecting it in the opposite direction of the global mean generates the inverse or anti-face [32]. In contrast with traditional linear measurements (e.g. distances), each direction or trait may affect multiple parts in the face at once, thus preserving the multivariate and multipartite nature of facial shape. New individuals can also be scored along the sib-shared traits, where the score continuously measures the presence or absence of that particular facial trait in all individuals (S4 Fig). To this end, we used two independently collected samples of unrelated individuals with European ancestry (EURO) originating from the United States (US, n = 4,680) and the United Kingdom (UK, n = 3,566) with genome-wide common variants available (S1 Table). First, SIB and EURO facial images were aligned in dense correspondence, ensuring that homology was established among the 7,160 quasi-landmarks [1,24]. Subsequently, univariate scores were generated by computing the MA between each trait vector and the EURO faces (S4 Fig), since this metric best captured facial trait similarity (S1 Fig).



Fig 3. Genetic loci associated with the sib-shared traits. (A) Number of sib-shared traits extracted per facial segment, corresponding to the number of sibling pairs that matched close to perfect within a given segment using the Mahalanobis angle. A total of 1,048 traits were extracted across all 63 segments, comprising 322 independent traits. The structure of the rosette plot corresponds to the polar dendrogram displaying the facial segments in Fig 2A. (B) Ideogram of the genetic loci that contribute to variation in the sib-shared traits, as identified by the association analysis of genome-wide common variants, depicted by circles and squares (i.e. overlapping and novel loci, respectively), and exome-wide low-frequency variants, depicted by triangles. For each locus, the color of the symbol represents the quadrant in which the top associated effect (i.e. lowest p-value) was observed.

https://doi.org/10.1371/journal.pgen.1009528.g003

Positive and negative scores corresponded to individuals whose identity vectors [27] were in similar ('face') or opposite ('anti-face') directions, respectively.

Genetic determination of sib-shared facial traits

Heritability. Narrow-sense heritability of the 1,048 sib-shared traits was estimated from single-nucleotide polymorphisms (SNPs) using GCTA [20,33]. Significance was evaluated according to the false discovery rate (FDR)-adjusted threshold (pFDR_{US} < 3.6×10^{-3} ; pFDR_{UK} < 2.2×10^{-3}), with significant SNP-based heritability estimates ranging from 0.17 to 0.42 in the US cohort and 0.24 to 0.53 in the UK cohort (S3 Table). Higher values were found on average for traits defined in the global face, nasal area and around the nasolabial folds (S5 Fig).

Low SNP-based heritability was observed for traits in small, locally defined areas around the cheeks, philtrum, forehead and chin.

Genetic association study. We conducted a genome-wide association scan (GWAS) on all 1,048 sib-shared traits in the US and UK cohorts separately and meta-analyzed the resulting p-values using inverse variance-weighting [22,34] (Figs <u>3B</u> and <u>S6</u>). We identified a total of 8,944 SNPs at 218 independent loci that reached the threshold for genome-wide significance ($p < 5 \times 10^{-8}$), of which 2,749 SNPs at 38 loci had p-values lower than the study-wide threshold adjusted for the effective number of independent tests ($p < 1.55 \times 10^{-10}$; <u>S4 Table</u>). The 218 lead SNPs individually explained on average 0.4% and up to 1.6% of the phenotypic variation for individual sib-shared traits. Together, they explained 5.0% to 10.2% for individual traits and approximately 6.6% of the total full-face variation. The LocusZoom plots of the 218 genome-wide significant findings and their associated facial effects are illustrated online [30].

A total of 548 sib-shared traits (52.3%, n = 219 independent traits [31]) were associated with at least one of the 218 genome-wide significant loci, providing 197 traits (18.8%, n = 116 independent traits) that reached the threshold for study-wide significance. Detected associations involved traits in a variety of facial segments, most of them representing variations in the nose (S7 Fig). Several loci showed significant associations with more than one facial region, while others had very localized effects (e.g. the tip of the nose) only [30]. Among the 218 loci, 109 (37 study-wide) overlapped with or were nearby (within 500 kb) the results of prior association studies of normal-range facial phenotypes, providing further support regarding their involvement in facial variation. In addition, we identified 109 loci (1 study-wide) not previously reported in related GWAS literature, some of which harbor putative craniofacial genes as implicated from human malformations (S4 Table).

In addition to the GWAS meta-analysis, we also studied genetic associations with low-frequency variants (MAF < 0.01) using a gene-based testing approach in a subset of the US cohort (PITT) with exome-wide data available (n = 1,906) (S1 Text). A total of 53 genes passed the exome-wide significance threshold (p < 3.94×10^{-6}), yet none surpassed the strict studywide significance level adjusted for the number of independent tests (p < 1.22×10^{-8}) (Fig 3B and S5 Table). Of these 53 exome-wide significant signals, five could be linked to genes associated with diverse craniofacial phenotypes (S1 Text).

Embryonic origin of craniofacial variation. We performed gene ontology analysis using GREAT [35] to study enrichment of biological processes and relevant phenotypes in the vicinity of the 218 genome-wide significant lead SNPs. A significant enrichment was observed for terms related to craniofacial development and morphogenesis (S8 Fig). In addition, our analysis implicated several limb-related processes and phenotypes. Common pathways to both facial and limb development are further evidenced by some 'cranio-digital' syndromes [36,37].

We next sought to identify tissues and cell-types enriched for active regulatory regions near the 218 genome-wide significant lead SNPs. We used ChIP-seq for histone H3 on lysine 27 (H3K27ac), a mark of active regulatory elements, from diverse cell types as described previously [25]. H3K27ac signals near the lead SNPs were most enriched in CNCCs, a transient, embryonic population of cells that give rise to most structures of the craniofacial complex [38,39]. Enrichment was also found for other embryonic craniofacial tissues at different stages of craniofacial development (within first 8 weeks of gestation), suggesting both an early embryonic origin and lasting signals through craniofacial development, respectively (Fig 4).

Genetic and phenotypic correlations. We explored both phenotypic and genetic correlations among the sib-shared traits using GCTA (S9 Fig and S6 Table) [20]. High absolute phenotypic and genetic correlations, within and between facial segments, were found together (S9 Fig: diagonal), consistent with the observation that many of the loci identified influence multiple aspects of facial morphology. In addition, environmental factors might also contribute to



Fig 4. Preferential activity in CNCCs and embryonic craniofacial tissues. Boxplots of the distribution of H3K27ac ChIP-seq signals in 20 kb regions around the 218 genome-wide significant lead SNPs in various adult, embryonic and in vitro–derived cell types. Samples corresponding to CNCCs (blue), embryonic craniofacial tissue (orange) and surface ectoderm (green) are highlighted.

https://doi.org/10.1371/journal.pgen.1009528.g004

the observed phenotypic correlations, i.e. for those pairs of traits that had little genetic overlap (S9 Fig: upper left corner).

We also investigated the genetic overlap between our derived facial traits and publicly available non-facial traits and diseases (n = 38; S7 Table) using cross-trait LDSC [21,40] (S2 Text). However, given the relatively modest sample size of the combined EURO cohort (n = 8,246), standard errors for genetic correlations were fairly large (mean = 0.27) and no significant values could be observed after adjustment for multiple testing (p < 4.09 x 10^{-6} ; S10 Fig and S8 Table).

Discussion

Over the past decade, a versatile toolbox of computational and statistical genetics methods and accompanying open-access software has been developed to investigate the genetic architecture of complex phenotypes. Examples include, but are not limited to, genome-wide association scans followed by fine mapping [41], rare variant mapping (e.g. burden tests) [23,42], and estimations of SNP-based heritability (e.g. linear mixed models) [20,33] as well as genetic correlations across multiple phenotypes (e.g. LD-score regression) [21,40]. Unfortunately, the deployment of these powerful tools onto the human face is hampered by the multivariate and multipartite nature of facial variability. To overcome this shortcoming, we developed a family-informed strategy for facial phenotyping that reduces 3D facial shape to univariate measurements in a supervised manner, though without resorting to arbitrary and subjective trait selection.

We set up a biometric identification experiment leading to the identification of phenotypically similar sibling pairs from which their overlapping facial traits were derived, each representing global or local aspects of facial morphology that were informative for accurately matching the siblings. Individuals from additional cohorts, independent of their genetic background, can then be scored on the sib-shared traits by measuring the angle between the new individual and the vector of the sib-shared trait, in a common coordinate system. Simultaneously, complex 3D facial structures are converted into univariate descriptors, allowing for the application of contemporary genomic tools. Although traditional univariate, anthropometric measures will remain a valuable resource because of their widespread use and simplicity, they fail to capture the full morphological complexity of 3D facial shape. In contrast, sib-shared traits defined in this work may affect multiple parts in the face at once, thereby preserving its multivariate and multipartite nature. Alternative strategies exist to reduce complex shape to univariate traits such as PCA, where each individual PC constitutes a unique facial trait. How-ever, PCA does not necessarily imply biological relevance, whereas the use of traditional epide-miological approaches (e.g. family-based designs) to drive facial phenotyping facilitates a focus on facial traits that are genetically determined [26,43].

In the biometric identification task, the MA as similarity measure correctly identified the greatest number of siblings from among their peers. Note that, although we use self-reported kinship, the presence of half-sibs or unrelated pairs will have little to no effect as they most likely will not be selected in the matching experiment. Two main observations were made: angular measurements outperformed distance measurements; and Mahalanobis metrics were more discriminative than Euclidean metrics. These results are in line with previous research that demonstrated the role of the angle between vectors, encoded as deviations from a global average, in the perception of facial similarity rather than physical distances [27,32,44]. Moreover, normalization by the standard deviation of each PC in the Mahalanobis space allowed for the reduction of large sources of variation while smaller sources were amplified, so that all types of variation captured by each PC contributed equally when similarity was evaluated [26,27]. O'Toole et al. [45] stated that lower PC dimensions (i.e. those associated with small eigenvalues) convey useful information for distinguishing faces from one another and hence guide recognition. An increased matching performance thus suggests that, besides identityspecific information, family-specific features are encoded in the lower dimensions as well. In view of these results, we used the MA as a metric to score individuals along a given trait direction, thereby neglecting the magnitude of vector. However, the inclusion of the magnitude via distance-based metrics may be valuable beyond the study of normal-range facial variation, for instance within in a clinical context since some dysmorphologies are observed at the extreme ends (distance-based) of the normal spectrum of variation [46,47].

Facial similarity between siblings was studied in the context of different global-to-local facial segments, determined by the structural variations in the EURO reference space. The EURO cohort was chosen as a reference because of its larger sample size in comparison with the SIB cohort, yielding a more robust segmentation of the face of which existing variations are more accurately described following PCA in combination with parallel analysis (S11 Fig). Furthermore, matching of siblings was independent from the selected reference space, either EURO- or SIB-based, when an equal number of dimensions was considered. In contrast, the lower number of PCs originally retained in the SIB reference space following parallel analysis reduced the matching performance, so that the selection of the EURO cohort as a reference is preferred (S11D Fig).

Although matching at the global, full-face level performed better than any individual local matcher, the highest identification rate was achieved through the integration of information across segments. Specifically, almost half of the sibling pairs (44.32%) could be matched within the top-1% when segment-wise MA-corrected features were combined. The improvement of multibiometric systems over unibiometric systems is well-known as they consolidate multiple sources of evidence compensating for the limitations in performance of individual matchers [48]. In other words, siblings will still be matched if they have some, but not all, features in common. This also corresponds to the expected degree of phenotypic resemblance between relatives [49] and highlights the added value of the global-to-local segmentation approach. In this work, we simply concatenated the feature sets of all segments branching from the full face into one single vector, thereby neglecting their correlation structure [50]. However, diverse feature selection methods and fusion techniques exist [48]. In automatic kinship verification

literature, accuracies of above 90% are achieved this way [51,52]. In these studies, researchers seek to determine the kin relationship given a pair of 2D images using different feature-based, metric learning and deep learning approaches [51-54]. However, maximizing the performance was not the main focus of this study. Instead, we aimed to identify facial traits that are genetically driven starting from known kin relationships. In this context, we sought to identify obviously matching sib-pairs.

In previous work, we used biometric authentication methods to establish multiple 'face-to-DNA' classifiers, each predicting DNA-encoded aspects (e.g. sex, genomic background, individual genetic loci) from facial shape in order to match given faces against a DNA profile [29]. Similarly, we can construct a 'face-to-face' classifier as an alternative to the current selection of phenotypically similar siblings based on rank-1% identification rates. For example, a classifier can be trained on the facial similarity between pairs of children for individual and combined segments. When applied to an independent test set of faces, the classifier outputs the probability that the similarity score provided was derived from a pair of siblings. Higher probability signifies greater phenotypic similarity, from which sib-shared traits can be derived. However, classification tasks require the data to be partitioned into proper training and test sets, which reduces the data sample size to work on for specific parts. For example, in such a scenario and in this work only the test data (typically a substantial lower percentage of the total dataset) is eligible for defining sib-shared traits as input to subsequent analyses. Therefore, the number of facial traits as input to the GWAS is very likely to be less, since the likelihood of closely resembling sib-pairs goes down with lower sample size.

Interestingly and affirmative of the proposed methodology, facial regions that were more often shared between siblings (e.g. chin, orbital and nasal region) coincide with regions of high heritability as described independently in previous work [19]. Similarly, average heritability estimated from SNPs was higher for traits defined in the global face and nasal region, consistent with the literature [19,55–57]. Low-to-moderate estimates for traits defined in the regions near the chin and forehead were observed, whereas greater heritability was estimated from twin and family data in these regions using a similar UK cohort [19,58]. This discrepancy could be attributed to the difference in study design (e.g. family or population-based), as SNP-based heritability only provides a lower bound of heritability that is tagged by common SNPs resulting from the genotyping and imputation efforts.

In a genome-wide meta-analysis, we observed 38 study-wide significant loci associated with normal-range facial morphology in individuals of European descent. An additional 180 loci surpassed the conventional genome-wide threshold of significance. Given that association signals close to this threshold are likely to be enriched for genuine signals [59], we report on the full list of 218 loci reaching genome-wide significance. The strongest, study-wide significant signals were found for loci that have been replicated multiple times by independent studies, both within and across different populations. Examples include the well-established genetic regions surrounding TBX15 [15,17,25,60], PAX3 [7,8,14,15,17,25,60], RUNX2 [9,14,17,25,60], SOX9 [15,17,25,60,61], PAX1 [9,10,15,25] among others. In addition, promising candidate genes were found near several of the genome-wide significant regions. For instance, the 17q24.1 region was associated with self-reported chin dimpling in the study of Pickrell et al. [60]. This same region showed an association in the current study (p-value = 3.71×10^{-8}), though the effect was located near the area surrounding the nostrils (trait 339, segment 8). The nearest gene, AXIN2, plays a critical role in craniofacial and axial skeleton development of mice [62,63]. The lead SNP rs8080680 is an eQTL of AXIN2 in blood (GTEx), and overlaps with promoter and enhancer epigenetic marks in many cell types [64]. An association with morphology of the upper lip (trait 763, segment 38) and nasal bridge (trait 376, segment 10) was found in the 9q22.31 region (p-value = 2.39×10^{-8}). Previous GWAS also identified

associations of the same region with aspects of nasal morphology [15,25,60]. The lead SNP rs4275276 is an intron variant of *ROR2*, in which mutations were found to cause a severe skeletal dysplasia known as Robinow syndrome-1 (OMIM 268310). Multiple dysmorphic facial features have been described in patients with Robinow syndrome-1 including frontal bossing, hypertelorism, long philtrum, triangular mouth and a broad nose [65], consistent with the associated effects observed in our GWAS. In a mouse model, *Ror2* has been shown to play a crucial role at multiple sites during organogenesis, providing a developmental basis for the distinct clinical features and anomalies described for Robinow syndrome [66,67].

In addition to the replicated findings, we also identified a total of 109 loci (1 study-wide) that did not overlap with the results of prior GWAS on normal-range facial variation. The 1q25.3 region showed a study-wide significant association (p-value = 6.12×10^{-11}) with downturning corners of the mouth (trait 609, segment 25). Among the genes within 500kb of the lead SNP rs6695444, CACNA1E is a potentially relevant candidate gene. Mutations in the CACNA1E cause developmental and epileptic encephalopathy (DEE; OMIM 618285), a severe and genetically heterogeneous neurodevelopmental disorder characterized by characterized by refractory infantile-onset seizures, severe hypotonia, and profound developmental impairment [68]. Although facial dysmorphism has been described for DEEs caused by mutations in different genes, the link with craniofacial morphology and DEE caused by mutations in CACNA1E is less clear. In the study of Helbig et al. [68], macrocephaly was often present in patients with CACNA1E mutations, but no other dysmorphic facial features could be observed. Among the novel genome-wide significant loci, some harbor genes known to play a role in craniofacial development or malformations. For example, we observed a signal near FOXE1 (rs113771540, p-value = 4.29×10^{-8}) associated with nose (trait 548, segment 21) and chin (trait 625, segments 6) morphology, and a peak spanning EPHA3 (rs73137393, p-value = 4.49×10^{-8}) showed an association in the buccal region (trait 422, segment 12). FOXE1 and EPHA3 are both orofacial clefting candidate genes [69,70] and have not been associated previously with normal-range facial features. Furthermore, several genes near the GWAS signals have been implicated in human congenital syndromes with craniofacial manifestations. For example, a genome-wide significant association with forehead prominence (trait 447, segment 14) was found in the 20p11.23 region (p-value = 4.78×10^{-8}). The lead SNP rs6136885 is an intronic variant of the RIN2 gene. Mutations in this gene were found to cause RIN2 syndrome, formerly known as macrocephaly, alopecia, cutis laxa and scoliosis (MACS) syndrome (OMIM 613075), which is a rare connective tissue disorder characterized by multiple facial and skeletal anomalies [71,72]. Macrocephaly has been described as a clinical feature in some patients, which also corresponds to the facial effect observed in our GWAS. The 8q23.3 region near *RAD21* showed an association (rs4876648, p-value = 3.21×10^{-8}) with variation in the tip of the nose (trait 224, segment 5). Mutations in RAD21 result in Cornelia de Lange syndrome (OMIM 614701), a developmental disorder characterized by mild intellectual disability and several facial dysmorphisms [73].

With reference to the study of White et al. [25], in which the same population cohort (EURO) was used, a considerable degree of overlap in the associated genetic loci was observed (n = 100 genome-wide, n = 37 study-wide significant), though both studies also identified a number of distinct regions. The difference between both approaches is that White et al. [25] ran a GWAS on multivariate phenotypes using canonical correlation analysis (CCA). Whereas CCA allows for an open-ended description of facial variation, that is, it extracts the facial effects most correlated with the genotypes, we here searched for specific traits of interest within each facial segment prior to the association scan. Although the percentage of variance explained by the sib-shared traits per segment was generally high (up to 96%), phenotypic variations present in the philtrum and chin regions were described to a lesser extent (S12 Fig).

Moreover, decomposition of the H3K27ac signals into those generated by overlapping and novel loci (S13 Fig) demonstrated that enrichment for multipotent, undifferentiated CNCCs, consistent with an earliest embryonic origin, was primarily driven by loci also identified by White et al. [25]. Interestingly, our non-overlapping SNPs showed preferential activity for other relevant embryonic craniofacial tissues that represent progressively later (though still within the first two months of gestation) timepoints in facial development and would thus contain greater amounts of CNCC-derived chondrocytes and osteoblasts, among other cell-types. These findings suggest that the extracted facial traits, which seem to segregate strongly in families, might represent features that form later in craniofacial development. Hence, the use of family-informed strategies alongside an open-ended approach may reveal additional insights into the genetic architecture of complex traits.

When focusing on the genetic correlation among sib-shared traits, we found substantial overlap of genetic variants contributing to variation in these traits. These correlation patterns were in line with the phenotypic correlation, as one could expect from the relationship between both for human traits [74]. Moreover, they also reflect the close embryological relationship of human facial variability. Genetic variation associated with facial morphology might also contribute to various other traits and disease. For example, distinct patterns of facial variation have been described in the literature for various neurodevelopmental and neuropsychiatric disorders with large genetic heterogeneity such as epilepsy [75], autism spectrum disorder [76], schizophrenia [77], and bipolar disorder [78]. Given the univariate nature of the sib-shared facial traits, cross-trait LD score regression (LDSC) can be applied to estimate the genetic overlap with non-facial traits, but much larger sample sizes are required to achieve adequate statistical power [21,40]. Therefore, at the current sample size (n = 8,246), standard errors were large (mean = 0.27) and no significant correlations could be observed (S2 Text).

SNP-based heritability and association signals suggest that common genetic variants contribute to variation in sibling-derived facial traits. As expected, higher SNP-based heritability estimates were found on average for traits that surpassed the threshold for genome-wide significance in our GWAS (two-sample t-test p-value = 4×10^{-4} (US) and 5×10^{-4} (UK)). However, there still is great overlap in the heritability range between traits that did or did not lead to significant findings when testing for genotype-phenotype associations (S14 Fig and S9 Table). This also limits our ability to further prioritize sib-shared facial traits in terms of their genetic determination in order to reduce the multiple testing burden. The lack of significant GWAS findings for a subset of traits shared by siblings (47.7%) might be attributed to a lack of power to detect variants with weak phenotypic effects, epistatic interactions among variants not picked up by GWAS, and common environmental factors shared by family members. Especially the latter is an important source of bias in family-based designs [49]. In addition, the familial occurrence of particular facial traits such as a square chin [79] suggests that facial features are determined by major gene effects in addition to polygenic effects. Since both common and rare variants can be passed on from parent to child, both can be the reason for siblings to look alike, but the current single SNP-based association approach only supports the identification of common variants. Therefore, we complemented our GWAS with an exomewide analysis of low-frequency variants, identifying 53 exome-wide significant associations of which eight could be linked to genes associated with diverse craniofacial phenotypes (S1 Text). However, none of the genes tested surpassed the threshold for study-wide significance at the current sample size in the PITT subcohort (n = 1,906; S5 Table).

In conclusion, we describe a data-driven strategy to facial phenotyping, supervised by the phenotypic resemblances between siblings. Traits of interest were then followed up with extensive qualitative analysis using diverse bioinformatics resources. We demonstrate that variability of the sib-shared traits is low to moderately heritable and identify a total of 218 genome-

wide (38 study-wide) significant loci associated with normal-range facial morphology. Not only do we provide additional support for numerous previously reported loci, we also identify 109 new genome-wide significant signals (1 study-wide), some of them harboring promising candidate genes as implicated from human craniofacial malformations. Moreover, our analyses indicate a preferential activity of the novel loci in embryonic craniofacial tissue compared with CNCCs, suggesting their action in further differentiated cell-types of the face including osteoblasts and chondrocytes, further complemented by their involvement in limb development as evidenced by gene ontology analysis. A number of follow-up analyses were conducted on the derived set of univariate sib-shared features, such as the analysis of low-frequency variants and genetic correlations using GCTA and LDSC (S1 and S2 Texts). However, only weak associations could be observed, which highlights the need for sample size to increase power. Multivariate methods have been proven to be extremely powerful in dissecting the genetic architecture of craniofacial variation through GWAS for even modest sample sizes [17,25]. Applied to the current sibling-based design, univariate investigations can be followed up with the segment-wise merging and combined analysis of traits derived from a single sib-pair in a multivariate setting. In line with our observation from the biometric matching of siblings using combined segments, we might expect a similar increase in performance and hence statistical power. With regard to potential follow-up studies, a continuing investment in the development of multivariate equivalents to already-existing tools will therefore be of great value to the field. In addition, focusing on specific phenotypes and families, in combination with sequencing technologies, will further enable the identification of rare variants and dominant patterns of inheritance. Such phenotypes can not only be derived from phenotypically similar sibling pairs, as proposed in this study, but extended pedigrees and faces from patients presented with a syndrome might be of great value. Finally, these findings should be followed-up by further replication efforts in larger samples in addition to functional studies in order to elucidate the biological mechanisms that control facial development, with numerous applications in the clinic and beyond.

Materials and methods

Ethics statement

Ethical approval was obtained at each recruitment site and all participants gave their written informed consent prior to participation. For individuals under 18 years of age, written consent was obtained from a parent or legal guardian. For the SIB sample, the following ethics approval was obtained: Ethics Committee Research UZ/KU Leuven (S56392: ML10285). For the PITT sample, the following local ethics approvals were obtained: Pittsburgh, PA (PITT IRB #PRO09060553 and #RB0405013); Seattle, WA (Seattle Children's IRB #12107); Houston, TX (UT Health Committee for the Protection of Human Subjects #HSC-DB-09-0508); and Iowa City, IA (University of Iowa Human Subjects Office IRB #200912764 and #200710721). For the PSU sample, the following local ethics approvals were obtained: Urbana-Champaign, IL (PSU IRB #13103); New York, NY (PSU IRB #45727); Cincinnati, OH (UC IRB #2015-3073); Twinsburg, OH (PSU IRB #2503); State College, PA (PSU IRB #44929 and #4320); Austin, TX (PSU IRB #44929); and San Antonio, TX (PSU IRB #1278). For the IUPUI sample, the following local ethics approval was obtained: Indianapolis, IN and Twinsburg, OH (IUPUI IRB #1409306349). For the UK sample, ethical approval for the study (Project B2261: "Exploring distinctive facial features and their association with known candidate variants") was obtained from the ALSPAC Ethics and Law Committee and the Local Research Ethics Committees. Informed consent for the use of data collected via questionnaires and clinics was obtained from participants following the recommendations of the ALSPAC Ethics and Law Committee

at the time. Consent for biological samples has been collected in accordance with the Human Tissue Act (2004).

Sample and recruitment

Our study included one family-based (SIB) and one population-based (EURO) cohort. The basic demographic features of both cohorts are provided in <u>S1 Table</u>. For the SIB cohort, data from 647 children from 358 nuclear families were obtained. Families were recruited through various media channels at the Center for Human Genetics (University Hospital of Leuven, Belgium) and at Technopolis, the Flemish Center for Science Communication, Belgium. 3D facial surface scans were available for all children and questionnaires providing information on demographic factors (e.g. sex, age, self-reported ancestry), general physical characteristics (e.g. height, weight), and family relationship were completed by a parent. Only full siblings of self-reported European ancestry, aged 5 to 15 years (median age = 9), were retained for analysis. A further reduction was done by excluding participants with missing data on any of the aforementioned variables and participants with poor quality images. The final study sample consisted of 424 children from 194 nuclear families, containing 273 unique sibling pairs (78 brother-brother, 79 sister-sister, 116 brother-sister; <u>S1 Table</u>). The number of siblings per family ranged from 2 to 5.

The EURO cohort (n = 8,246) included 3D facial images and genotype data of four independent samples, three originating from the US and one from the UK, each composed of unrelated individuals of European ancestry [25] (S1 Table). The US dataset included samples obtained through different studies at the University of Pittsburgh (PITT), Pennsylvania State University (PSU), and Indiana University-Purdue University Indianapolis (IUPUI). Information on demographic factors (e.g. sex, age, self-reported ancestry) and general physical characteristics (e.g. height, weight) were available for all US participants. Individuals were excluded if they reported a personal or family history of any birth defect or syndrome affecting the head or face, a personal history of any significant facial trauma or facial surgery, or any medical condition that might alter the structure of the face. A further reduction was done by excluding participants with missing genotype data, missing covariates, or 3D image artifacts. Lastly, only individuals of European descent were retained, which we identified through projections into a principal component (PC) space constructed using the 1000G Phase 3 dataset (see below). The final PITT sample included 1,906 unrelated individuals (aged 3 to 40 years, median age = 23) from the 3D Facial Norms repository [80]. The final PSU sample consisted of 1,990 unrelated individuals (aged 18 to 88 years, median age = 24). For the IUPUI sample a total of 784 unrelated individuals (aged 7 to 78 years, median age = 19) met all quality-control criteria.

The UK dataset was derived from the Avon Longitudinal Study of Parents and Children (ALSPAC), a UK-based birth cohort study [81,82]. A total of 14,541 pregnant women with an expected delivery date between 1 April 1991 and 31 December 1992, were initially recruited. Extensive information and biological samples have been collected from these mothers and their offspring at various time points, of which details can be found on the study website through a fully searchable data dictionary (http://www.bris.ac.uk/alspac/researchers/data-access/data-dictionary/). Here, 3D facial images, genotype data and self-reported information on age, sex, height, and weight were available for 3,566 unrelated adolescents (aged 14 to 17 years, median age = 15).

Genotyping, quality control, imputation and population structure

PITT participants were genotyped on the Illumina OmniExpress + Exome v1.2 array, plus 4,322 investigator-chosen SNPs included to capture variation in specific regions of interest

based on previous studies of the genetics of facial variation. For the PSU sample, participants were either genotyped on the Illumina Human Hp200c1 BeadChip or on the 23andMe v3 and v4 arrays. Participants from the IUPUI sample were genotyped on the Illumina's Infinium Multi-Ethnic Global-8 v1 array. Standard data cleaning and quality assurance procedures were performed based on the GRCh37 (hg19) genome assembly using PLINK 1.9 [83]. Specifically, samples were evaluated for concordance of genetic and reported sex, evidence of chromosomal aberrations, genotype call rate (--mind 0.1), and batch effects. SNPs were evaluated for call rate (--geno 0.1), Mendelian errors, deviation from Hardy-Weinberg genotype proportions (--hwe 0.01), and sex differences in allele frequency and heterozygosity.

Genotypes in the PITT, PSU and IUPUI samples, separately, were imputed to the 1000 Genomes Project Phase 3 reference panel [84]. First, pre-phasing of haplotypes was performed in SHAPEIT2 [85], and imputation of nearly 40 million variants was performed using the Sanger Imputation Server [86] with the Positional Burrows-Wheeler Transform (PBWT) pipe-line [87]. SNP-level (INFO score < 0.8) and genotype per participant-level (genotype probability < 0.9) filters were used to omit poorly-imputed variants. Finally, a single US cohort (n = 4,680) was obtained by merging the subsamples and filtering the SNPs based on missingness across individuals (--geno 0.5), minor allele frequency (--maf 0.01), and Hardy-Weinberg equilibrium (p < 1 x 10⁻⁶), ultimately resulting in 7,417,619 SNPs for analysis [25].

European individuals in the US cohort were selected using principal component analysis (PCA) of approximately 450,000 SNPs, after excluding from the imputed data all indels, multiallelic SNPs, and SNPs with low MAF (≤ 0.1), and SNPs in linkage disequilibrium (50 bp window, 5 bp step size, 0.2 correlation threshold) from the 1000G Phase 3 dataset. A k-nearest neighbor algorithm was used to assign a 1000G population label to each US participant, and those with the 1000G European label of CEU, TSI, FIN, GBR, and IBS were selected for analysis only [25].

For the UK dataset, genotype information was obtained directly from the ALSPAC database. Because restrictions are in place against merging the ALSPAC genotypes with any other genotypes, these were held separately during the analysis. UK participants were genotyped on the Illumina HumanHap550 quad chip platform by Sample Logistics and Genotyping Facilities at the Wellcome Sanger Institute and LabCorp (Laboratory Corporation of America), supported by 23andMe. Genotypes were subjected to standard quality control methods. SNPs were evaluated for minor allele frequency (removed if < 0.01), call rate (removed if < 0.95), and deviation from Hardy-Weinberg genotype proportions (removed if $p < 5 \times 10^{-7}$). Individuals were excluded on the basis of gender mismatches, minimal or excessive heterozygosity, disproportionate levels of individual missingness (removed if > 3%), and insufficient sample replication (removed if IBD < 0.8). Only individuals of European descent, compared to the HapMap II dataset by way of multidimensional scaling analysis, were kept for imputation. SHAPEIT2 [85] was used for pre-phasing of haplotypes and imputation against the 1000 Genomes Phase 1 reference panel (Version 3) [88] was performed using IMPUTE2 [89]. The final UK sample consisted of 3,566 individuals with 8,629,873 SNPs for analysis [25].

3D facial imaging and shape variables

3D facial image acquisition. 3D facial surface images were acquired from all participants using two digital stereophotogrammetry systems and one laser scanning system, applying standard facial image acquisition protocols [2]. For the PITT sample, 3D images were obtained with the 3dMDface camera system (3dMD, Atlanta, GA). Image data of the IUPUI sample were acquired using the Vectra H1 (Canfield Scientific) system, and both 3dMDface and Vectra H1 systems were used for the SIB and PSU samples. For the UK dataset, 3D facial images

were captured with the Konica Minolta Vivid (VI900) laser scanners (Konica Minolta Sensing Europe Company, Milton Keynes, United Kingdom).

Spatially dense facial quasi-landmarking and quality control. Dense surface registration was performed in Matlab 2017b using the MeshMonk toolbox [1]. In essence, standardized spatially dense quasi-landmark configurations were established by non-rigidly mapping a symmetric template composed of thousands of points (n = 7,160) onto the images [24]. All datasets (SIB, US, UK) were processed separately with the same template, thereby creating homology between the three study cohorts.

Imaging and mapping errors, presented as outlier faces, were detected by a two-step quality control. First, deviations of a face from the global, within-cohort, average were converted to z-scores and images with a z-score larger than 2 were manually checked [17]. A second exclusion criteria involved the percentage of correspondence outliers reported by the MeshMonk toolbox [1], arising from the presence of artifacts such as holes in the facial surface. Similarly, images with scores reflecting a large proportion of outliers were manually checked and removed if necessary.

Standardized and quality-controlled images were aligned in dense correspondence (position, orientation, and size) by generalized Procrustes analysis (GPA). This was done for original and reflected configurations combined, with the latter constructed by changing the sign of the x-coordinate [90]. The average of an original and its reflected configuration constitutes the symmetric component of facial variation, while the difference between the two constitutes the asymmetric component of facial variation. Because faces display bilateral symmetry, aspects of symmetry and asymmetry are preferably considered separately when examining facial shape [91]. Although patterns of asymmetry may be informative, in this work we concentrate on the symmetric component only.

Global-to-local facial segmentations. A single EURO cohort was obtained by combining and GPA-aligning the US and UK datasets. We used this large cohort [25] as a reference to perform global-to-local segmentations of the face. Because all individuals, including the SIB cohort, were processed with the same facial template, SIB participants could be projected into the corresponding EURO shape space.

For the three datasets (SIB, US, UK) separately, symmetrized facial shapes were first adjusted for the confounders of sex, age, age², height, weight, facial size and camera system [92] in a partial least-squares regression (PLSR, function plsregress from Matlab 2017b). An additional correction for population structure was performed for both US and UK datasets by including the first four genetic PCA axes (i.e. ancestry axes) in the PLSR model. Next, global-to-local segmentations (Fig 2A) of adjusted faces were performed by grouping vertices that are strongly correlated, characterized by Escoufier's RV coefficient [93], in a hierarchical spectral clustering approach [17]. Per segment, facial shapes were aligned through GPA, followed by PCA combined with parallel analysis to adequately capture significant facial variations with fewer PCs. In this way, a shape space was established for each segment independently while their integration was preserved through the hierarchical construction. The same segmentation was applied to the SIB dataset and participants were brought into a common global-to-local coordinate system through projections of the SIB shapes into the EURO reference space.

Data-driven selection of facial phenotypes by matching siblings

Measures of facial similarity and sibling matching. Individual faces can be described as single points or vectors within a multidimensional space, whose dimensions characterize facial variation [27,32] (S4 Fig). Image similarity between faces within the PCA space was measured in four ways: Euclidean distance (ED), Mahalanobis distance (MD), Euclidean angle (EA), and

Mahalanobis angle (MA). The ED is the simple straightforward linear distance between two points, whereas MD is the variation-normalized version of ED [26]. Normalization was achieved by dividing the PC scores of the SIB cohort by the standard deviation of each PC. Angle-based measurements refer to the cosine distance between two points, treated as vectors, and can be computed from the cosine of the included angle between those vectors. In sum, the lower the distance or angle between two points, the greater the similarity between those individuals.

The ability of the similarity measures introduced above to match siblings was tested using a biometric identification setup for individual and combined segments. The multidimensional PC scores constituted the facial features of each participant for each facial segment. Shape information was combined by concatenating the Euclidean or Mahalanobis-corrected feature sets of multiple segments (segments 1–63) into a single feature vector per participant, similar to feature level fusion in multibiometric systems [48], after which the distance and angle were computed.

In a one-to-many setup, the identification task aimed to identify the one true sibling among a gallery of faces. The gallery was built from the combination of the corresponding sibling and all non-relatives present in the SIB cohort. When multiple (>2) siblings per family existed, the matching experiment was repeated for each pair provided that all other relatives were omitted. Per segment, a total of 546 experiments (2 x 273 unique pairs, as both members of a sibling pair were used as the query image in a separate round) were performed with gallery sizes varying from 417 to 423. In essence, similarity between the query image and all possible candidates in the gallery was measured and ranked in decreasing order from the most likely to least likely candidate. A final score was assigned per experiment reflecting the position of the true sibling in the sorted gallery list. The performance was assessed via cumulative match characteristic (CMC) curves, which plots the cumulative rank-k% identification rate for different values of *k*. High identification rates and a steep slope of the curve at higher ranks (i.e. low rank values) indicated better performance [29].

Selection of sib-shared traits. Per segment, sibling pairs were selected if they matched close to perfect (within the top-1%) in the identification task using MA, as this measure performed best in the sibling matching, and when they scored within the lowest 2.5 percentile for the measured similarity. The second constraint was imposed to differentiate between the matching on similarity and distinctiveness (e.g. atypical features) [44] as the latter might form discrete clusters in shape space. In addition, matching behavior had to be symmetric. That is, both selection requirements should have been met regardless of which member of the sibling pair was chosen as the query image. In total, 218 sibling pairs met all selection criteria in at least one facial segment. We defined the final phenotype as the average facial shape of a particular sibling pair within a given segment for which good matching behavior was observed, represented by a vector with reference to the overall EURO mean shape. A total of 1,048 traits were defined across all sibling pairs and facial segments, each of them highlighting the facial features, either global or locally defined, that were informative for accurately matching the siblings. The 1,048 sib-shared traits comprised 322 independent traits [31] due to the hierarchical and overlapping construction of the facial segments and the presence of multiple sibling pairs per nuclear family.

Conversion of sib-shared traits to univariate scores and genetic analyses

Conversion of sib-shared traits. Complex shape transformations, encoded by the sibshared traits, were converted into simple univariate phenotypes so that well-established toolboxes could be used to assess genetic contributions to the traits. This was achieved by scoring unrelated individuals along the vectors that make up the sib-shared traits. Scores were generated by computing the MA between the trait vector and all EURO individuals, as this measure best captured facial similarity in the matching of siblings, and represented the variation of the sib-shared traits in the EURO cohort (S4 Fig). Individuals whose corresponding vectors had a similar orientation (small angle), independent of their magnitude, received scores close to 1 ([0 1] interval). In contrast, negative scores ([-1 0] interval) were assigned to individuals whose vectors were in opposite directions (large angle).

Estimation of SNP-based heritability. For the US and UK datasets separately, SNP-based heritability (h_{SNP}^2) of the sib-shared traits was estimated using GREML implemented in the GCTA package [20,94]. We fitted a linear mixed model with two variance components, including the full genetic relationship matrix between individuals estimated from the SNPs, and the residual variance. The proportion of the total variance explained by all SNPs provided a measure of the narrow-sense heritability. We determined the false discovery rate (FDR) p-value threshold at $p < 3.6 \times 10^{-3}$ and $p < 2.2 \times 10^{-3}$ in the US and UK cohorts, respectively.

Genome-wide association meta-analysis. Similar to the adjustment of facial phenotypes, imputed genotypes were corrected for the effects of sex, age, age², height, weight, facial size, camera system [92], and the first four genetic PCA axes using PLSR. We fitted a linear regression model (function regstats from Matlab 2017b) under an additive genetic model to test for associations between each SNP and each of the sibling-derived phenotypes. For SNPs on the X chromosome, males were coded 0/2 to be on the same scale as 0/1/2 females. Analyses were performed separately for the US and UK cohorts and the resulting p-values were combined in a meta-analysis via inverse variance-weighting [22,33].

We used the conventional threshold of $p < 5 \ge 10^{-8}$ to claim genome-wide statistical significance. Given the expected number of correlated traits due to the hierarchical design of the facial segmentation and the existence of multiple sibling pairs within a single family, the threshold for study-wide significance was determined at $p < 1.55 \ge 10^{-10}$ (i.e. $p < 5 \ge 10^{-8}$ divided by 322), corresponding to an adjustment for the number of independent phenotypes estimated from the eigenvalues of the phenotypic correlation matrix [31].

Because of the large number of signals considered, we used the relatively automated peak detection criterion based on genomic position solely. SNPs that reached the genome-wide threshold were grouped using a 1Mb window. For each region, the lead SNP was defined as the SNP with the lowest p-value for any of the derived traits, resulting in a total of 221 peaks. These peaks were followed up with examination of patterns of linkage disequilibrium (LD) [95], leading to the identification of three regions where association signals were spanning a very large region in the genome. These cases were subsequently merged, refining our results to 218 lead SNPs, all below the genome-wide threshold.

Genes 500 kb up- and downstream of the lead SNPs were identified using the Table Browser of the UCSC Genome Browser [96]. We first investigated whether these genes had any craniofacial relevance by searching the PubMed and OMIM [97] repositories and by looking at the overlap with existing facial GWAS literature. Here, overlap was determined by considering the same +/- 500 kb window around each lead SNP (similar to the definition of peaks) and by investigating LD between the identified and previously reported lead SNPs using LDlink [98]. When no biological plausible candidates were identified, we used FUMA [99] to assign the most likely candidate gene(s) using preset parameters. For each of the lead SNPs, biological functions were annotated using GREAT [35].

Cell-type-specific enhancer enrichment. Chromatin activity in the vicinity of the lead SNPs was quantified using H3K27ac ChIP-seq signals from approximately 100 different cell and tissue types, including human CNCCs [100], fetal and adult osteoblasts [101–103], mesen-chymal stem cell-derived chondrocytes [101], dissected embryonic craniofacial tissues [104],

and iPS-derived surface ectoderm [105], as described in detail by White et al. [25]. As part of this study, we added data from in-vitro-derived surface ectoderm [105]. To compare H3K27ac signals between cell-types in an unbiased manner, we divided the genome into 20 kb windows, and calculated H3K27ac reads per million from each aligned read (bam or tagAlign) file in each window using bedtools coverage. After quantile normalization (using the normalize. quantiles function from the preprocessCore package), we selected the windows containing each of the lead SNPs, random SNPs matched for minor allele frequency and distance to the lead SNPs using SNPsnap [106] and Crohn's disease-associated SNPs from the NCBI-EBI GWAS catalog [107], the latter serving as a positive control.

Phenotypic and genetic correlations. We estimated the Pearson's phenotypic correlation coefficients among the sib-shared traits. To estimate the genetic correlation between sib-shared traits that is tagged by SNPs, we conducted a bivariate GREML analysis [108,109] of corrected phenotypes in the US and UK datasets separately, as implemented in GCTA [20]. We determined the FDR p-value threshold at $p < 1.07 \times 10^{-5}$ and $p < 1.00 \times 10^{-5}$ in the US and UK cohorts, respectively.

Supporting information

S1 Fig. Biometric identification results. Cumulative match characteristic curves of (A) the full-face (segment 1) and combined segments for the different similarity measures and (B) individual local matchers and combined segments using the Mahalanobis angle. The diagonal line represents random performance. ED, Euclidean distance; EA, Euclidean angle; MD, Mahalanobis distance; MA, Mahalanobis angle. (TIF)

S2 Fig. Biometric identification versus gallery size. Rank-k and rank-k% identification rates for varying gallery sizes based on the full-face matching of siblings using the Mahalanobis angle. The experiment was repeated 1,000 times, with mean identification rates represented by the solid (rank-k) and dashed (rank-k%) lines, and the minimum and maximum performance indicated by the shaded area. For rank-k% matchings, results are plotted for gallery sizes of 100 and above. In case of rank-1% identification, results are valid only for multiples of 100 as ranks cannot have non-integer values, explaining the decreasing/increasing pattern observed. (TIF)

S3 Fig. Biometric identification results for different sex-based groups. Cumulative match characteristic curves of individual local matchers for the three sex-based groups (n = 78 brother-brother, n = 79 sister-sister, n = 116 brother-sister). Facial similarity was determined using the Mahalanobis angle. The diagonal line represents random performance. (TIF)

S4 Fig. Supervised scoring of individuals onto sib-shared traits. (A) Illustration of a multidimensional facial space using PCA, and (B) supervised scoring of individuals onto a specific shape direction coding for the sib-shared trait, depicted by the blue nasal shape. Positive scores indicate the presence of facial features similar to those shared by siblings, while negative scores correspond to features opposite to the sibling pair (left y-axis). The score distribution for all EURO participants is plotted on top of the histogram (right y-axis). (TIF)

S5 Fig. Mean heritability of the sib-shared traits per facial segment. Mean phenotypic variance explained by commons SNPs in the US cohort and UK cohort. SNP-based heritability (h^2_{SNP}) of the 1,048 sib-shared traits was estimated using GCTA and average values per

segment are plotted on top of each node. The structure of the rosette plot corresponds to the polar dendrogram displaying the facial segments in Fig 2A. (TIF)

S6 Fig. Manhattan plot of genetic variants associated with the sib-shared traits. Combined Manhattan plot of the sib-shared traits, highlighting the novel and overlapping loci in red and blue, respectively. Per SNP, the lowest meta-analysis p-value across all 1,048 traits is plotted. The solid horizontal line represents the genome-wide significance threshold ($p < 5 \ge 10^{-8}$) and the dashed horizontal line represents the study-wide threshold ($p < 1.55 \ge 10^{-10}$). (TIF)

S7 Fig. Genome-wide significant associations per facial segment. (A) Number of genomewide significant loci that showed an association with one (or multiple) of the sib-shared traits, defined within a particular segment. (B) Proportion of sib-shared traits defined per facial segment (%) that showed an association with at least one of the 218 genome-wide significant loci. The structure of the rosette plot corresponds to the polar dendrogram displaying the facial segments in Fig 2A.



S8 Fig. GREAT analysis of the 218 genome-wide significant loci. Top 15 gene ontology enrichment of biological process GO terms, human phenotypes and mouse phenotypes. Plotted is the binomial test FDR (blue) and binomial enrichment (orange). (TIF)

S9 Fig. Genetic and phenotypic correlations among sib-shared traits. Relationship between genetic correlation p-values (x-axis) and the phenotypic correlation (y-axis) in the US and UK cohort. Pairwise correlations between traits that were derived from the same family were excluded. R_p, phenotypic correlation; R_g, genetic correlation. (TIF)

S10 Fig. Genetic correlations between sib-shared traits and non-facial traits and diseases. Pairwise correlations between facial and non-facial traits that reached nominal significance (p < 0.05), computed using cross-trait LDSC. Facial traits are sorted per quadrant, corresponding to the polar dendrogram displaying the facial segments in Fig 2A. (TIF)

S11 Fig. EURO- and SIB-based reference space. Hierarchical facial segmentation and number of significant principal components determined by parallel analysis in the (A) EURO (n = 8,246) and (B) SIB (n = 424) cohort. (C) Number of significant components retained by parallel analysis in varying, randomly generated subsets of the EURO cohort. (D) Cumulative match characteristic curves for full-face matchings (segment 1) of siblings based on the Mahalanobis angle in a EURO-based (solid line) and SIB-based (dashed line) reference space. In the 'SIB70' space (dotted line), the number of dimensions is equal to the original EURO reference space. ED, Euclidean distance; EA, Euclidean angle; MD, Mahalanobis distance; MA, Mahalanobis angle.

(TIF)

S12 Fig. Variance explained by the sib-shared traits per facial segment. The amount of variation explained by the sib-shared traits expressed as percentage for each facial segment. The structure of the rosette plot corresponds to the polar dendrogram displaying the facial segments in Fig 2A.

(TIF)

S13 Fig. Preferential activity in CNCCs and embryonic craniofacial tissues. Shown are the boxplots of the distribution of H3K27ac ChIP-seq signals in 20 kb regions around the (A) 218 lead SNPs, (B) 100 overlapping SNPs and (C) 118 non-overlapping SNPs in various adult, embryonic and in vitro–derived cell types. Overlap was determined with reference to the study of White et al. [25], who utilized the same European study cohort in a multivariate GWAS. Samples corresponding to CNCCs (blue), embryonic craniofacial tissue (orange) and surface ectoderm (green) are highlighted. (TIF)

S14 Fig. SNP-based heritability and genome-wide associations of the sib-shared facial traits. Link between SNP-based heritability and (A) study-wide ('SW') and (B) genome-wide ('GW') significance of sib-shared traits in the GWAS meta-analysis. Traits that didn't reach statistical significance in the GWAS are coded as '0'; traits that were associated with at least one of the identified loci are coded as '> = 1'. The two-sample t-test p-value is plotted on top of each panel, with significant values indicated in bold. (TIF)

S1 Table. Summary of the study cohort. (XLSX)

S2 Table. Biometric identification results for individual and combined segments. R1, rank-1% identification rate; R10, rank-10% identification rate; R20, rank-20% identification rate; ED, Euclidean distance; EA, Euclidean angle; MD, Mahalanobis distance; MA, Mahalanobis angle. (XLSX)

S3 Table. SNP-based heritability of the sib-shared traits. Significant heritability estimates are indicated in bold. h^2_{SNP} , SNP-based heritability; SE, standard error. (XLSX)

S4 Table. Genome-wide association of common variants with sib-shared facial traits in the discovery samples (US, UK) and combined meta-analysis. The column 'Best Trait' represents the index (1–1,048) of the sib-shared trait in which the lowest meta p-value ('Best P-value') was found, together with the corresponding segment and quadrant in which the original trait was defined ('Best Segment' and 'Best Quadrant'). MAF, minor allele frequency; SE, standard error.

(XLSX)

S5 Table. Gene-based association of low-frequency variants with sib-shared facial traits in the PITT subsample. NumVar, number of variants tested per gene. (XLSX)

S6 Table. Genetic correlations among the sib-shared traits. Pairwise genetic correlations in the US cohort (lower/left triangle) and UK cohort (upper/right triangle). Standard errors are displayed between brackets. (XLSX)

S7 Table. Overview of publicly available traits and diseases tested for correlations with the sib-shared traits.

(XLSX)

S8 Table. Genetic correlations between sib-shared traits and non-facial traits and diseases. Rg, genetic correlation; SE, standard error. (XLSX) **S9** Table. SNP-based heritability and genome-wide associations of the sib-shared facial **traits**. Statistical significance is indicated in bold.

(XLSX)

S1 Text. Exome-wide low frequency variant analysis. (PDF)

S2 Text. Genetic correlation between sib-shared traits and non-facial traits and diseases. (PDF)

Acknowledgments

We are extremely grateful to all the families who took part in this study, the midwives for their help in recruiting them, and the whole ALSPAC team, which includes interviewers, computer and laboratory technicians, clerical workers, research scientists, volunteers, managers, receptionists and nurses. We are also very grateful to all of the SIB and US participants and lab members for generously donating their time, and Technopolis Belgium for allowing the use of their facilities for this research.

Author Contributions

Conceptualization: Hanne Hoskens, Greet Hens, Hilde Peeters, Peter Claes.

- **Data curation:** Hanne Hoskens, Myoung Keun Lee, Karlijne Indencleef, Julie D. White, Jiarui Li, Susan Walsh, Peter Claes.
- **Formal analysis:** Hanne Hoskens, Dongjing Liu, Sahin Naqvi, Myoung Keun Lee, Ryan J. Eller, Karlijne Indencleef, Julie D. White, Jiarui Li.
- **Funding acquisition:** Greet Hens, Joanna Wysocka, Susan Walsh, Stephen Richmond, Mark D. Shriver, John R. Shaffer, Hilde Peeters, Seth M. Weinberg, Peter Claes.
- **Investigation:** Hanne Hoskens, Dongjing Liu, Sahin Naqvi, Myoung Keun Lee, Ryan J. Eller, Karlijne Indencleef, Maarten H. D. Larmuseau, Susan Walsh, Stephen Richmond, Mark D. Shriver, John R. Shaffer, Seth M. Weinberg, Peter Claes.
- Methodology: Hanne Hoskens, Peter Claes.
- **Resources:** Susan Walsh, Stephen Richmond, Mark D. Shriver, John R. Shaffer, Seth M. Weinberg, Peter Claes.
- Software: Hanne Hoskens, Dongjing Liu, Sahin Naqvi, Peter Claes.
- Supervision: Greet Hens, Joanna Wysocka, Susan Walsh, Mark D. Shriver, John R. Shaffer, Hilde Peeters, Seth M. Weinberg, Peter Claes.
- Visualization: Hanne Hoskens, Dongjing Liu, Sahin Naqvi, Myoung Keun Lee, Peter Claes.
- Writing original draft: Hanne Hoskens, Dongjing Liu, Sahin Naqvi, Peter Claes.
- Writing review & editing: Hanne Hoskens, Dongjing Liu, Sahin Naqvi, Myoung Keun Lee, Ryan J. Eller, Karlijne Indencleef, Julie D. White, Jiarui Li, Maarten H. D. Larmuseau, Greet Hens, Joanna Wysocka, Susan Walsh, Stephen Richmond, Mark D. Shriver, John R. Shaffer, Hilde Peeters, Seth M. Weinberg, Peter Claes.

References

 White JD, Ortega-Castrillón A, Matthews H, Zaidi AA, Ekrami O, Snyders J, et al. MeshMonk: Opensource large-scale intensive 3D phenotyping. Sci Rep. 2019; 9(1):6085. https://doi.org/10.1038/ s41598-019-42533-y PMID: 30988365

- Heike CL, Upson K, Stuhaug E, Weinberg SM. 3D digital stereophotogrammetry: a practical guide to facial image acquisition. Head Face Med. 2010; 6(18). https://doi.org/10.1186/1746-160X-6-18 PMID: 20667081
- Zaidi AA, Mattern BC, Claes P, McEcoy B, Hughes C, Shriver MD. Investigating the case of human nose shape and climate adaptation. Gibson G, editor. PLOS Genet. 2017; 13(3):e1006616. <u>https://doi.org/10.1371/journal.pgen.1006616</u> PMID: 28301464
- Kau CH, Richmond S, Incrapera A, English J, Xia JJ. Three-dimensional surface acquisition systems for the study of facial morphology and their application to maxillofacial surgery. Int J Med Robotics Comput Assist Surg. 2007; 3(2):97–110.
- Jheon AH, Oberoi S, Solem RC, Kapila S. Moving towards precision orthodontics: an Evolving paradigm shift in the planning and delivery of customized orthodontic therapy. Orthod Craniofac Res. 2017; 20:106–113. https://doi.org/10.1111/ocr.12171 PMID: 28643930
- Hammond P. The use of 3D face shape modelling in dysmorphology. Arch Dis Child. 2007; 92 (12):1120–1126. https://doi.org/10.1136/adc.2006.103507 PMID: 18032641
- Paternoster L, Zhurov AI, Toma AM, Kemp JP, St. Pourcain B, Timpson NJ, et al. Genome-wide association study of three-dimensional facial morphology identifies a variant in *PAX3* associated with nasion position. Am J Hum Genet. 2012; 90(3):478–485. https://doi.org/10.1016/j.ajhg.2011.12.021 PMID: 22341974
- Liu F, van der Lijn F, Schurmann C, Zhu G, Chakravarty MM, Hysi PG, et al. A genome-wide association study identifies five loci influencing facial morphology in Europeans. PLOS Genet. 2012; 8(9): e1002932. https://doi.org/10.1371/journal.pgen.1002932 PMID: 23028347
- Adhikari K, Fuentes-Guajardo M, Quinto-Sánchez M, Mendoza-Revilla J, Camilo Chacón-Duque J, Acuña-Alonzo V, et al. A genome-wide association scan implicates DCHS2, RUNX2, GLI3, PAX1 and EDAR in human facial variation. Nat Commun. 2016; 7:11616. https://doi.org/10.1038/ncomms11616 PMID: 27193062
- Shaffer JR, Orlova E, Lee MK, Leslie EJ, Raffensperger ZD, Heike CL, et al. Genome-wide association study reveals multiple loci influencing normal human facial morphology. PLOS Genet. 2016; 12 (8):e1006149. https://doi.org/10.1371/journal.pgen.1006149 PMID: 27560520
- Cole JB, Manyama M, Kimwaga E, Mathayo J, Larson JR, Liberton DK, et al. Genomewide association study of African children identifies association of SCHIP1 and PDE8A with facial size and shape. Barsh GS, editor. PLOS Genet. 2016; 12(8):e1006174. <u>https://doi.org/10.1371/journal.pgen.1006174</u> PMID: 27560698
- Howe LJ, Lee MK, Sharp GC, Smith GD, Pourcain BS, Shaffer JR, et al. Investigating the shared genetics of non-syndromic cleft lip/palate and facial morphology. PLOS Genet. 2018; 14(8):e1007501. https://doi.org/10.1371/journal.pgen.1007501 PMID: 30067744
- Qiao L, Yang Y, Fu P, Hu S, Zhou H, Peng S, et al. Genome-wide variants of Eurasian facial shape differentiation and a prospective model of DNA based face prediction. J Genet Genomics. 2018; 45 (8):419–432. https://doi.org/10.1016/j.jqg.2018.07.009 PMID: 30174134
- Li Y, Zhao W, Li D, Tao X, Xiong Z, Liu J, et al. EDAR, LYPLAL1, PRDM16, PAX3, DKK1, TNFSF12, CACNA2D3, and SUPT3H gene variants influence facial morphology in a Eurasian population. Hum Genet. 2019; 138:681–689. https://doi.org/10.1007/s00439-019-02023-7 PMID: 31025105
- Xiong Z, Dankova G, Howe LJ, Lee MK, Hysi PG, de Jong MA, et al. Novel genetic loci affecting facial shape variation in humans. eLife. 2019; 8:e49898. https://doi.org/10.7554/eLife.49898 PMID: 31763980
- Crouch DJM, Winney B, Koppen WP, Christmas WJ, Hutnik K, Day T, et al. Genetics of the human face: Identification of large-effect single gene variants. Proc Nat Acad Sci. 2018; 115(4):e676–685. https://doi.org/10.1073/pnas.1708207114 PMID: 29301965
- Claes P, Roosenboom J, White JD, Swigut T, Sero D, Li J, et al. Genome-wide mapping of global-tolocal genetic effects on human facial shape. Nat Genet. 2018; 50(3):414–423. https://doi.org/10.1038/ s41588-018-0057-4 PMID: 29459680
- Johannsdottir B, Thorarinsson F, Thordarson A, Magnusson TE. Heritability of craniofacial characteristics between parents and offspring estimated from lateral cephalograms. Am J Orthod Dentofacial Orthop. 2005; 127(2):200–207. https://doi.org/10.1016/j.ajodo.2004.07.033 PMID: 15750539
- Hoskens H, Li J, Indencleef K, Gors D, Larmuseau MHD, Richmond S, et al. Spatially dense 3D facial heritability and modules of co-heritability in a father-offspring design. Front Genet. 2018; 9:554. <u>https:// doi.org/10.3389/fgene.2018.00554</u> PMID: 30510565
- Yang J, Lee SH, Goddard ME, Visscher PM. GCTA: a tool for genome-wide complex trait analysis. Am J Hum Genet. 2011; 88(1):76–82. https://doi.org/10.1016/j.ajhg.2010.11.011 PMID: 21167468

- Bulik-Sullivan BK, Loh P-R, Finucane HK, Ripke S, Yang J, Schizophrenia Working Group of the Psychiatric Genomics Consortium, et al. LD Score regression distinguishes confounding from polygenicity in genome-wide association studies. Nat Genet. 2015; 47(3):291–295. <u>https://doi.org/10.1038/ng</u>. 3211 PMID: 25642630
- 22. Willer CJ, Li Y, Abecasis GR. METAL: fast and efficient meta-analysis of genomewide association scans. Bioinformatics. 2010; 26(17):2190–2191. https://doi.org/10.1093/bioinformatics/btq340 PMID: 20616382
- Lee S, Emond MJ, Bamshad MJ, Barnes KC, Rieder MJ, Nickerson DA, et al. Optimal unified approach for rare-variant association testing with application to small-sample case-control wholeexome sequencing studies. Am J Hum Genet. 2012; 91(2):224–237. https://doi.org/10.1016/j.ajhg. 2012.06.007 PMID: 22863193
- Claes P, Walters M, Clement J. Improved facial outcome assessment using a 3D anthropometric mask. Int J Oral Maxillofac Surg. 2012; 41(3):324–330. https://doi.org/10.1016/j.ijom.2011.10.019 PMID: 22103995
- White JD, Indencleef K, Naqvi S, Eller RJ, Hoskens H, Roosenboom J, et al. Insights into the genetic architecture of the human face. Nat Genet. 2021; 53(1):45–53. <u>https://doi.org/10.1038/s41588-020-00741-7</u> PMID: 33288918
- Aeria G, Claes P, Vandermeulen D, Clement JG. Targeting specific facial variation for different identification tasks. Forensic Sci Int. 2010 Sep; 201(1–3):118–124. <u>https://doi.org/10.1016/j.forsciint.2010</u>. 03.005 PMID: 20359838
- Hill H, Claes P, Corcoran M, Walters M, Johnston A, Clement JG. How different is different? Criterion and sensitivity in face-space. Front Psychol. 2011; 2:41. <u>https://doi.org/10.3389/fpsyg.2011.00041</u> PMID: 21738516
- 28. Jain AK, Flynn P, Ross AA. Handbook of biometrics. New York: Springer, New York; 2008. 556 p.
- Sero D, Zaidi A, Li J, White JD, Zarzar TBG, Marazita ML, et al. Facial recognition from DNA using face-to-DNA classifiers. Nat Commun. 2019; 10(1):2557. https://doi.org/10.1038/s41467-019-10617-y PMID: 31186421
- Hoskens H. FigShare 3D facial phenotyping by biometric sibling matching used in contemporary genomic methodologies. Available from: https://doi.org/10.6084/m9.figshare.c.5231540
- **31.** Li J, Ji L. Adjusting multiple testing in multilocus analyses using the eigenvalues of a correlation matrix. Heredity. 2005; 95(3):221–227. https://doi.org/10.1038/sj.hdy.6800717 PMID: 16077740
- 32. Griffin HJ, McOwan PW, Johnston A. Relative faces: encoding of family resemblance relative to gender means in face space. J Vis. 2011; 11(12):8. https://doi.org/10.1167/11.12.8 PMID: 22003253
- Evans LM, Tahmasbi R, Vrieze SI, Abecasis GR, Das S, Gazal S, et al. Comparison of methods that use whole genome data to estimate the heritability and genetic architecture of complex traits. Nat Genet. 2018; 50(5):737–745. https://doi.org/10.1038/s41588-018-0108-x PMID: 29700474
- Begum F, Ghosh D, Tseng GC, Feingold E. Comprehensive literature review and statistical considerations for GWAS meta-analysis. Nucleic Acids Res. 2012; 40(9):3777–3784. <u>https://doi.org/10.1093/ nar/gkr1255</u> PMID: 22241776
- McLean CY, Bristor D, Hiller M, Clarke SL, Schaar BT, Lowe CB, et al. GREAT improves functional interpretation of cis-regulatory regions. Nat Biotechnol. 2010; 28(5):495–501. https://doi.org/10.1038/ nbt.1630 PMID: 20436461
- Winter RM. What's in a face? Nat Genet. 1996; 12:124–129. <u>https://doi.org/10.1038/ng0296-124</u> PMID: 8563748
- Rosano A, Botto LD, Olney RS, Khoury MJ, Ritvanen A, Goujard J, et al. Limb defects associated with major congenital anomalies: clinical and epidemiological study from the International Clearinghouse for Birth Defects Monitoring Systems. Am J Med Genet. 2000; 93(2):110–116. https://doi.org/10.1002/ 1096-8628(20000717)93:2<110::aid-ajmg6>3.0.co;2-9 PMID: 10869112
- Cordero DR, Brugmann S, Chu Y, Bajpai R, Jame M, Helms JA. Cranial neural crest cells on the move: Their roles in craniofacial development. Am J Med Genet. 2011; 155(2):270–279. <u>https://doi.org/10.1002/ajmg.a.33702</u> PMID: 21271641
- Bronner ME, LeDouarin NM. Development and evolution of the neural crest: an overview. Dev Biol. 2012; 366(1):2–9. https://doi.org/10.1016/j.ydbio.2011.12.042 PMID: 22230617
- Bulik-Sullivan B, Finucane HK, Anttila V, Gusev A, Day FR, Loh P-R, et al. An atlas of genetic correlations across human diseases and traits. Nat Genet. 2015; 47(11):1236–1241. https://doi.org/10.1038/ ng.3406 PMID: 26414676
- Schaid DJ, Chen W, Larson NB. From genome-wide associations to candidate causal variants by statistical fine-mapping. Nat Rev Genet. 2018; 19(8):491–504. https://doi.org/10.1038/s41576-018-0016z PMID: 29844615

- Zhan X, Hu Y, Li B, Abecasis GR, Liu DJ. RVTESTS: an efficient and comprehensive tool for rare variant association analysis using sequence data. Bioinformatics. 2016; 32(9):1423–1426. https://doi.org/ 10.1093/bioinformatics/btw079 PMID: 27153000
- 43. Mitteroecker P, Gunz P. Advances in geometric morphometrics. Evol Biol. 2009; 36(2):235–247.
- 44. Ross DA, Hancock PJB, Lewis MB. Changing faces: direction is important. Vis Cogn. 2010; 18(1):67–81.
- O'Toole AJ, Deffenbacher KA, Valentin D, Abdi H. Low-dimensional representation of faces in higher dimensions of the face space. J Opt Soc Am A. 1993; 10(3):405.
- Attanasio C, Nord AS, Zhu Y, Blow MJ, Li Z, Liberton DK, et al. Fine tuning of craniofacial morphology by distant-acting enhancers. Science. 2013; 342(6157):1241006. <u>https://doi.org/10.1126/science.</u> 1241006 PMID: 24159046
- Boehringer S, van der Lijn F, Liu F, Günther M, Sinigerova S, Nowak S, et al. Genetic determination of human facial morphology: links between cleft-lips and normal variation. Eur J Hum Genet. 2011; 19 (11):1192–1197. https://doi.org/10.1038/ejhg.2011.110 PMID: 21694738
- **48.** Ross AA, Nandakumar K, Jain AK. Handbook of multibiometrics. Boston, MA: Springer, Boston; 2006.
- **49.** Falconer DS, Mackay TFC. Introduction to quantitative genetics. 4th ed. Harlow: Longman Group Ltd; 1996.
- Jain A, Nandakumar K, Ross A. Score normalization in multimodal biometric systems. Pattern Recognit. 2005; 38(12):2270–2285.
- Wu X, Boutellaa E, Feng X, Hadid A. Kinship verification from faces: Methods, databases and challenges. In: 2016 IEEE International Conference on Signal Processing, Communications and Computing (ICSPCC). Hong Kong, China: IEEE; 2016:1–6.
- 52. Bordallo Lopez M, Hadid A, Boutellaa E, Goncalves J, Kostakos V, Hosio S. Kinship verification from facial images and videos: human versus machine. Mach Vis Appl. 2018; 29(5):873–890.
- 53. Vieira TF, Bottino A, Laurentini A, De Simone M. Detecting siblings in image pairs. Vis Comput. 2014; 30(12):1333–1345.
- 54. Fang R, Tang KD, Snavely N, Chen T. Towards computational models of kinship verification. In: 2010 IEEE International Conference on Image Processing, Hong Kong, 2010:1577–1580.
- 55. Tsagkrasoulis D, Hysi P, Spector T, Montana G. Heritability maps of human face morphology through large-scale automated three-dimensional phenotyping. Sci Rep. 2017; 7:45885. <u>https://doi.org/10.1038/srep45885 PMID: 28422179</u>
- 56. Cole JB, Manyama M, Larson JR, Liberton DK, Ferrara TM, Riccardi SL, et al. Human facial shape and size heritability and genetic correlations. Genetics. 2017; 205(2):967–978. <u>https://doi.org/10. 1534/genetics.116.193185 PMID: 27974501</u>
- Djordjevic J, Zhurov AI, Richmond S, Visigen Consortium. Genetic and environmental contributions to facial morphological variation: a 3D population-based twin study. Cai T, editor. PLOS ONE. 2016; 11 (9):e0162250. https://doi.org/10.1371/journal.pone.0162250 PMID: 27584156
- Djordjevic J, Jadallah M, Zhurov AI, Toma AM, Richmond S. Three-dimensional analysis of facial shape and symmetry in twins using laser surface scanning. Orthod Craniofac Res. 2013; 16(3):146– 160. https://doi.org/10.1111/ocr.12012 PMID: 23323545
- Panagiotou OA, Ioannidis JPA, for the Genome-Wide Significance Project. What should the genomewide significance threshold be? Empirical replication of borderline genetic associations. Int J Epidemiol. 2012; 41(1):273–286. https://doi.org/10.1093/ije/dyr178 PMID: 22253303
- Pickrell JK, Berisa T, Liu JZ, Ségurel L, Tung JY, Hinds DA. Detection and interpretation of shared genetic influences on 42 human traits. Nat Genet. 2016; 48(7):709–717. <u>https://doi.org/10.1038/ng.</u> 3570 PMID: 27182965
- Cha S, Lim JE, Park AY, Do J-H, Lee SW, Shin C, et al. Identification of five novel genetic loci related to facial morphology by genome-wide association studies. BMC Genomics. 2018; 19(1):481. <u>https:// doi.org/10.1186/s12864-018-4865-9 PMID: 29921221</u>
- McGee-Lawrence ME, Li X, Bledsoe KL, Wu H, Hawse JR, Subramaniam M, et al. Runx2 protein represses Axin2 expression in osteoblasts and is required for craniosynostosis in Axin2-deficient Mice. J Biol Chem. 2013; 288(8):5291–5302. https://doi.org/10.1074/jbc.M112.414995 PMID: 23300083
- Li C, Lan Y, Krumlauf R, Jiang R. Modulating Wnt signaling rescues palate morphogenesis in Pax9 mutant mice. J Dent Res. 2017; 96(11):1273–1281. https://doi.org/10.1177/0022034517719865 PMID: 28692808

- Ward LD, Kellis M. HaploReg: a resource for exploring chromatin states, conservation, and regulatory motif alterations within sets of genetically linked variants. Nucleic Acids Res. 2012; 40(D1):D930–934.
- van Bokhoven H, Celli J, Kayserili H, van Beusekom E, Balci S, Brussel W, et al. Mutation of the gene encoding the ROR2 tyrosine kinase causes autosomal recessive Robinow syndrome. Nat Genet. 2000; 25(4):423–426. https://doi.org/10.1038/78113 PMID: 10932187
- 66. Schwabe GC, Trepczik B, Suring K, Brieske N, Tucker AS, Sharpe PT, et al. Ror2 knockout mouse as a model for the developmental pathology of autosomal recessive Robinow syndrome. Dev Dynam. 2004; 229:400–410. https://doi.org/10.1002/dvdy.10466 PMID: 14745966
- Oishi I, Suzuki H, Onishi N, Takada R, Kani S, Ohkawara B, et al. The receptor tyrosine kinase Ror2 is involved in non-canonical Wnt5a/JNK signalling pathway: Role of Ror2 in Wnt5a signalling pathway. Genes to Cells. 2003; 8(7):645–654. https://doi.org/10.1046/j.1365-2443.2003.00662.x PMID: 12839624
- Helbig KL, Lauerer RJ, Bahr JC, Souza IA, Myers CT, Uysal B, et al. De novo pathogenic variants in CACNA1E cause developmental and epileptic encephalopathy with contractures, macrocephaly, and dyskinesias. Am J Hum Genet. 2018; 103(5):666–678. https://doi.org/10.1016/j.ajhg.2018.09.006 PMID: 30343943
- Agrawal P, Wang M, Kim S, Lewis AE, Bush JO. Embryonic expression of EphA receptor genes in mice supports their candidacy for involvement in cleft lip and palate. Dev Dynam. 2014; 243(11):1470– 1476. https://doi.org/10.1002/dvdy.24170 PMID: 25073978
- 70. Leslie EJ, Carlson JC, Shaffer JR, Butali A, Buxó CJ, Castilla EE, et al. Genome-wide meta-analyses of nonsyndromic orofacial clefts identify novel associations between FOXE1 and all orofacial clefts, and TP63 and cleft lip with or without cleft palate. Hum Genet. 2017; 136(3):275–286. https://doi.org/ 10.1007/s00439-016-1754-7 PMID: 28054174
- Basel-Vanagaite L, Sarig O, Hershkovitz D, Fuchs-Telem D, Rapaport D, Gat A, et al. RIN2 deficiency results in macrocephaly, alopecia, cutis laxa, and scoliosis: MACS Syndrome. Am J Hum Genet. 2009; 85(2):254–263. https://doi.org/10.1016/j.ajhg.2009.07.001 PMID: 19631308
- Rosato S, Syx D, Ivanovski I, Pollazzon M, Santodirocco D, De Marco L, et al. RIN2 syndrome: Expanding the clinical phenotype. Am J Med Genet. 2016; 170(9):2408–2415. <u>https://doi.org/10.1002/ajmg.a.37789</u> PMID: 27277385
- Krab LC, Marcos-Alcalde I, Assaf M, Balasubramanian M, Andersen JB, Bisgaard A-M, et al. Delineation of phenotypes and genotypes related to cohesin structural protein RAD21. Hum Genet. 2020; 139 (5):575–592. https://doi.org/10.1007/s00439-020-02138-2 PMID: 32193685
- 74. Sodini SM, Kemper KE, Wray NR, Trzaskowski M. Comparison of genotypic and phenotypic correlations: Cheverud's conjecture in humans. Genetics. 2018; 209(3):941–948. <u>https://doi.org/10.1534/</u> genetics.117.300630 PMID: 29739817
- Kasperavičiūtė D, Catarino CB, Chinthapalli K, Clayton LMS, Thom M, Martinian L, et al. Uncovering genomic causes of co-morbidity in epilepsy: gene-driven phenotypic characterization of rare microdeletions. PLOS ONE. 2011; 6(8):e23182. <u>https://doi.org/10.1371/journal.pone.0023182</u> PMID: 21858020
- 76. Aldridge K, George ID, Cole KK, Austin JR, Takahashi TN, Duan Y, et al. Facial phenotypes in subgroups of prepubertal boys with autism spectrum disorders are correlated with clinical phenotypes. Mol Autism. 2011; 2(1):15. https://doi.org/10.1186/2040-2392-2-15 PMID: 21999758
- 77. Buckley PF, Dean D, Bookstein FL, Han S, Yerukhimovich M, Min K-J, et al. A three-dimensional morphometric study of craniofacial shape in schizophrenia. Am J Psychiatry. 2005; 162(3):606–608. https://doi.org/10.1176/appi.ajp.162.3.606 PMID: 15741481
- Hennessy RJ, Baldwin PA, Browne DJ, Kinsella A, Waddington JL. Frontonasal dysmorphology in bipolar disorder by 3D laser surface imaging and geometric morphometrics: Comparisons with schizophrenia. Schizophrenia Research. 2010; 122(1–3):63–71. <u>https://doi.org/10.1016/j.schres.2010.05</u>. 001 PMID: 20554158
- 79. Vilas R, Ceballos FC, Al-Soufi L, González-García R, Moreno C, Moreno M, et al. Is the "Habsburg jaw" related to inbreeding? Ann Hum Biol. 2019; 46(7–8):553–561. <u>https://doi.org/10.1080/03014460. 2019.1687752</u> PMID: <u>31786955</u>
- Weinberg SM, Raffensperger ZD, Kesterke MJ, Heike CL, Cunningham ML, Hecht JT, et al. The 3D Facial Norms database: Part 1. A web-based craniofacial anthropometric and image repository for the clinical and research community. Cleft Palate Craniofac J. 2016; 53(6):185–197. https://doi.org/10. 1597/15-199 PMID: 26492185
- Boyd A, Golding J, Macleod J, Lawlor DA, Fraser A, Henderson J, et al. Cohort profile: the 'Children of the 90s'—the index offspring of the Avon Longitudinal Study of Parents and Children. Int J Epidemiol. 2013; 42(1):111–27. https://doi.org/10.1093/ije/dys064 PMID: 22507743

- Fraser A, Macdonald-Wallis C, Tilling K, Boyd A, Golding J, Davey Smith G, et al. Cohort profile: the Avon Longitudinal Study of Parents and Children: ALSPAC mothers cohort. Int J Epidemiol. 2013; 42 (1):97–110. https://doi.org/10.1093/ije/dys066 PMID: 22507742
- Chang CC, Chow CC, Tellier LC, Vattikuti S, Purcell SM, Lee JJ. Second-generation PLINK: Rising to the challenge of larger and richer datasets. GigaSci. 2015; 4(1):7. <u>https://doi.org/10.1186/s13742-015-0047-8 PMID: 25722852</u>
- Auton A, Abecasis GR, Altshuler DM, Durbin RM, Bentley DR, Chakravarti A, et al. A global reference for human genetic variation. Nature. 2015; 526(7571):68–74. https://doi.org/10.1038/nature15393
 PMID: 26432245
- Delaneau O, Zagury J-F, Marchini J. Improved whole-chromosome phasing for disease and population genetic studies. Nat Methods. 2013; 10(1):5–6. https://doi.org/10.1038/nmeth.2307 PMID: 23269371
- McCarthy S, Das S, Kretzschmar W, Delaneau O, Wood AR, Teumer A, et al. A reference panel of 64,976 haplotypes for genotype imputation. Nat Genet. 2016; 48(10):1279–1283. <u>https://doi.org/10. 1038/ng.3643 PMID: 27548312</u>
- Durbin R. Efficient haplotype matching and storage using the positional Burrows-Wheeler transform (PBWT). Bioinformatics. 2014; 30(9):1266–1272. https://doi.org/10.1093/bioinformatics/btu014 PMID: 24413527
- McVean GA, Altshuler DM, Durbin RM, Abecasis GR, Bentley DR, Chakravarti A, et al. An integrated map of genetic variation from 1,092 human genomes. Nature. 2012; 491(7422):56–65. https://doi.org/ 10.1038/nature11632 PMID: 23128226
- Howie B, Marchini J, Stephens M. Genotype imputation with thousands of genomes. G3 Genes, Genomes, Genetics. 2011; 1(6):457–470.
- 90. Claes P, Walters M, Vandermeulen D, Clement JG. Spatially-dense 3D facial asymmetry assessment in both typical and disordered growth. J Anat. 2011; 219(4):444–455. https://doi.org/10.1111/j.1469-7580.2011.01411.x PMID: 21740426
- 91. Claes P, Walters M, Shriver MD, Puts D, Gibson G, Clement J, et al. Sexual dimorphism in multiple aspects of 3D facial symmetry and asymmetry defined by spatially dense geometric morphometrics: Spatially dense sexual dimorphism in 3D facial shape. J Anat. 2012; 221(2):97–114. <u>https://doi.org/10.1111/j.1469-7580.2012.01528.x PMID: 22702244</u>
- White JD, Ortega-Castrillon A, Virgo C, Indencleef K, Hoskens H, Shriver MD, et al. Sources of variation in the 3dMDface and Vectra H1 3D facial imaging systems. Sci Rep. 2020; 10(1):4443. https://doi. org/10.1038/s41598-020-61333-3 PMID: 32157192
- **93.** Robert P, Escoufier Y. A unifying tool for linear multivariate statistical methods: the RV-coefficient. Appl Stat. 1976; 25(3):257.
- Yang J, Benyamin B, McEvoy BP, Gordon S, Henders AK, Nyholt DR, et al. Common SNPs explain a large proportion of the heritability for human height. Nat Genet. 2010; 42(7):565–569. <u>https://doi.org/ 10.1038/ng.608 PMID: 20562875</u>
- **95.** Pruim RJ, Welch RP, Sanna S, Teslovich TM, Chines PS, Gliedt TP, et al. LocusZoom: regional visualization of genome-wide association scan results. Bioinformatics. 2010; 26(18):2336–2337. <u>https://doi.org/10.1093/bioinformatics/btq419</u> PMID: 20634204
- Karolchik D. The UCSC Table Browser data retrieval tool. Nucleic Acids Res. 2004; 32(90001):D493– 496. https://doi.org/10.1093/nar/gkh103 PMID: 14681465
- Hamosh A. Online Mendelian Inheritance in Man (OMIM), a knowledgebase of human genes and genetic disorders. Nucleic Acids Res. 2004; 33(Database issue):D514–517.
- Machiela MJ, Chanock SJ. LDlink: a web-based application for exploring population-specific haplotype structure and linking correlated alleles of possible functional variants. Bioinformatics. 2015; 31 (21):3555–3557. https://doi.org/10.1093/bioinformatics/btv402 PMID: 26139635
- 99. Watanabe K, Taskesen E, van Bochoven A, Posthuma D. Functional mapping and annotation of genetic associations with FUMA. Nat Commun. 2017; 8(1):1826. https://doi.org/10.1038/s41467-017-01261-5 PMID: 29184056
- 100. Prescott SL, Srinivasan R, Marchetto MC, Grishina I, Narvaiza I, Selleri L, et al. Enhancer divergence and cis-regulatory evolution in the human and chimp neural crest. Cell. 2015; 163(1):68–83. <u>https:// doi.org/10.1016/j.cell.2015.08.036</u> PMID: 26365491
- 101. Kundaje A, Meuleman W, Ernst J, Bilenky M, Yen A, Heravi-Moussavi A, et al. Integrative analysis of 111 reference human epigenomes. Nature. 2015; 518(7539):317–330. <u>https://doi.org/10.1038/nature14248</u> PMID: 25693563

- 102. Baumgart SJ, Najafova Z, Hossan T, Xie W, Nagarajan S, Kari V, et al. CHD1 regulates cell fate determination by activation of differentiation-induced genes. Nucleic Acids Res. 2017; 45(13):7722–7735. https://doi.org/10.1093/nar/gkx377 PMID: 28475736
- 103. Najafova Z, Tirado-Magallanes R, Subramaniam M, Hossan T, Schmidt G, Nagarajan S, et al. BRD4 localization to lineage-specific enhancers is associated with a distinct transcription factor repertoire. Nucleic Acids Res. 2017; 45(1):127–141. https://doi.org/10.1093/nar/gkw826 PMID: 27651452
- 104. Wilderman A, VanOudenhove J, Kron J, Noonan JP, Cotney J. High-Resolution Epigenomic Atlas of Human Embryonic Craniofacial Development. Cell Rep. 2018; 23(5):1581–1597. <u>https://doi.org/10. 1016/j.celrep.2018.03.129 PMID: 29719267</u>
- 105. Pattison JM, Melo SP, Piekos SN, Torkelson JL, Bashkirova E, Mumbach MR, et al. Retinoic acid and BMP4 cooperate with p63 to alter chromatin dynamics during surface epithelial commitment. Nat Genet. 2018; 50(12):1658–1665. https://doi.org/10.1038/s41588-018-0263-0 PMID: 30397335
- 106. Pers TH, Timshel P, Hirschhorn JN. SNPsnap: a web-based tool for identification and annotation of matched SNPs. Bioinformatics. 2015; 31(3):418–420. https://doi.org/10.1093/bioinformatics/btu655 PMID: 25316677
- 107. Buniello A, MacArthur JAL, Cerezo M, Harris LW, Hayhurst J, Malangone C, et al. The NHGRI-EBI GWAS Catalog of published genome-wide association studies, targeted arrays and summary statistics 2019. Nucleic Acids Res. 2019; 47(D1):D1005–D1012. <u>https://doi.org/10.1093/nar/gky1120</u> PMID: 30445434
- Adams HHH, Hibar DP, Chouraki V, Stein JL, Nyquist PA, Rentería ME, et al. Novel genetic loci underlying human intracranial volume identified through genome-wide association. Nat Neurosci. 2016; 19 (12):1569–1582. https://doi.org/10.1038/nn.4398 PMID: 27694991
- 109. Lee SH, Yang J, Goddard ME, Visscher PM, Wray NR. Estimation of pleiotropy between complex diseases using single-nucleotide polymorphism-derived genomic relationships and restricted maximum likelihood. Bioinformatics. 2012; 28(19):2540–2542. <u>https://doi.org/10.1093/bioinformatics/bts474</u> PMID: 22843982